Author's response to reviews

Title: Impact of a Parent-Child Sexual Communication Campaign: Results from a Controlled Efficacy Trial of Parents

Authors:

Kevin C Davis (kcdavis@rti.org)
Jonathan L Blitstein (jblitstein@rti.org)
W Douglas Evans (Doug.Evans@gwumc.edu)
Kian Kamyab (kkamyab@rti.org)

Version: 2 Date: 6 July 2010

Author's response to reviews:

Dear Editor:

I am pleased to re-submit our revised paper “Impact of a Parent-Child Sexual Communication Campaign: Results from a Controlled Efficacy Trial of Parents” to BMC Reproductive Health. On behalf of the study coauthors, I would like to thank the reviewers for their helpful feedback. We have addressed each of the major compulsory and minor essential revisions in detail. In the attached “Response to Reviewer Comments” document, we provide a point-by-point explanation of how each comment was addressed in the revised manuscript. We hope you agree our revisions adequately address the reviewer comments. If any additional questions arise or anything needs further clarification in our responses, please do not hesitate to contact me.

Sincerely,

Kevin C. Davis, M.A.
Senior Research Economist
RTI International
3040 Cornwallis Rd
Research Triangle Park, NC 27709
Phone: 919-541-5801
Fax: 919-541-6683
E-mail: kcdavis@rti.org